**Author’s response to reviews**

**Title:** Superior vena cava syndrome in a dialysis patient: a case report.

**Authors:**

Uwe Goettmann (uwe.goettmann@umm.de)
Maliha Sadick (maliha.sadick@umm.de)
Kathrin Kleinhuber (kathrin.kleinhuber@umm.de)
Urs Benck (urs.benck@umm.de)
Kurt Huck (kurt.huck@umm.de)
Bernhard K. Krämer (bernhard.kraemer@umm.de)
Rainer Birck (rainer.birck@umm.de)

**Version:** 3  **Date:** 10 April 2012

**Author’s response to reviews:**

Re-Submission MS 1589210607651360

Dear Prof. Kidd,

Thank you very much for giving us the opportunity to submit a revised version of our case report entitled “Central vein stenosis in a dialysis patient: a case report” to your journal.

As requested by the associate editor we improved figure 2 and marked the vessels.

We apologize for the unclear marking of the figures. As figures 1-4 belong together and should appear together they are now integrated into one figure. The numbers of the following figures in the upload menu now correspond to the numbers in the manuscript.

We hope that the improved manuscript adds valuable information to your readership is now acceptable for publication in the “Journal of Medical Case Reports”.

Sincerely yours

Uwe Göttmann
M.D., Ph.D.